\begin{table}%t4
\rule{0cm}{2cm}
\caption{Elemental abundances in the sample stars.\label{tableabu}}
%\centerline
$${
\begin{tabular}{lr@{$~\pm~$}lr@{$~\pm~$}lr@{$~\pm~$}lr@{$~\pm~$}lr@{$~\pm~$}lr@{$~\pm~$}lr@{$~\pm~$}lr@{$~\pm~$}lr@{$~\pm~$}lr@{$~\pm~$}l}
\hline\hline
\noalign{\smallskip}
Element & \multicolumn{2}{c}{Sun$^{a}$} & 
\multicolumn{2}{c}{$\eta~$Leo} & 
\multicolumn{2}{c}{HD~111613} &
\multicolumn{2}{c}{HD~92207} & 
\multicolumn{2}{c}{$\beta$~Ori} & 
\multicolumn{2}{c}{Gal AI$^{b}$} & 
\multicolumn{2}{c}{Gal BV$^{c}$} & 
\multicolumn{2}{c}{Gal BV$^{d}$} &
\multicolumn{2}{c}{Gal BV$^{e}$} &
\multicolumn{2}{c}{Gal BV$^{f}$}\\
\hline
\ion{He}{i} & 10.99 & 0.02 & {\em 11.18} & {\em 0.04}~(14) & {\em 11.07} & {\em 0.05}~(10) & {\em 11.14} & {\em 0.04}~(10) & {\em 11.19} & {\em 0.04}~(15) & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & {\em 11.13} & {\em 0.22} & {\em 11.05} & {\em 0.10} & \multicolumn{2}{c}{\ldots}\\
\ion{C}{i} & 8.52 & 0.06 & {\em 7.94} & {\em 0.10}~(4) & {\em 8.10} & {\em 0.07}~(2) & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & {\em 8.14} & {\em 0.13} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots}\\
\ion{C}{ii} & 8.52 & 0.06 & {\em 8.10} & {\em 0.09}~(3) & {\em 8.24} & {\em 0.06}~(3) & \multicolumn{2}{c}{{\em 8.33}~(1)} & {\em 8.15} & {\em 0.05}~(3) & \multicolumn{2}{c}{\ldots} & {\em 8.27} & {\em 0.13} &{\em 8.20} & {\em 0.10} & {\em 8.22} & {\em 0.15} & 7.87 & 0.16\\
\ion{N}{i} & 7.92 & 0.06 & {\em 8.41} & {\em 0.09}~(20) & {\em 8.40} & {\em 0.10}~(16) & {\em 8.25} & {\em 0.04}~(11) & {\em 8.50} & {\em 0.07}~(11) & {\em 8.37} & {\em 0.21$^*$} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots}\\
\ion{N}{ii} & 7.92 & 0.06 & \multicolumn{2}{c}{{\em 8.32}~(1)} & \multicolumn{2}{c}{{\em 8.36}~(1)} & \multicolumn{2}{c}{{\em 8.28}~(1)} & {\em 8.51} & {\em 0.06}~(16) & \multicolumn{2}{c}{\ldots} & {\em 7.63} & {\em 0.15} & {\em 7.75} & {\em 0.27} & {\em 7.78} & {\em 0.27} & 7.90 & 0.22\\
\ion{O}{i} & 8.83 & 0.06 & {\em 8.78} & {\em 0.05}~(13) & {\em 8.70} & {\em 0.04}~(9) & {\em 8.79} & {\em 0.07}~(6) & {\em 8.78} & {\em 0.04}~(6) & 8.77 & 0.12 & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots}\\
\ion{O}{ii} & 8.83 & 0.06 & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & {\em 8.83} & {\em 0.03}~(5) & \multicolumn{2}{c}{\ldots} & {\em 8.55} & {\em 0.14} &{\em 8.64} & {\em 0.20} & {\em 8.52} & {\em 0.16} & 8.89 & 0.14\\
\ion{Ne}{i} & 8.08 & 0.06 & 8.39 & 0.07~(7) & 8.42 & 0.06~(5) & 8.50 & 0.14~(7) & 8.40 & 0.08~(9) & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & 8.10 & 0.05$^{**}$ & \multicolumn{2}{c}{\ldots}\\
\ion{Mg}{i} & 7.58 & 0.01 & {\em 7.52} & {\em 0.08}~(7) & {\em 7.46} & {\em 0.04}~(4) & {\em 7.60} & {\em 0.04}~(4) & \multicolumn{2}{c}{\ldots} & 7.48 & 0.17 & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots}\\
\ion{Mg}{ii} & 7.58 & 0.01 & {\em 7.53} & {\em 0.04}~(12) & {\em 7.43} & {\em 0.04}~(6)& {\em 7.40} & {\em 0.05}~(5) & {\em 7.42} & {\em 0.02}~(4) & 7.46 & 0.17 & {\em 7.42} & {\em 0.22} & {\em 7.59} & {\em 0.22} &  {\em 7.38} & {\em 0.12} & 7.70 & 0.34\\
\ion{Al}{i} & 6.49 & 0.01 & 6.11 & 0.06~(2) & 6.06 & 0.06~(2) & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots}\\
\ion{Al}{ii} & 6.49 & 0.01 & 6.39 & 0.17~(5) & 6.55 & 0.27~(6) & 6.38 & 0.38~(6) & 6.14 & 0.08~(4) & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots}\\
\ion{Al}{iii} & 6.49 & 0.01 & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & 7.00 & 0.38~(4) & \multicolumn{2}{c}{\ldots} & {\em 6.08} & {\em 0.12} & {\em 6.21} & {\em 0.19} & {\em 6.12} & {\em 0.18} & 6.31 & 0.15\\
\ion{Si}{ii} & 7.56 & 0.01 & 7.58 & 0.19~(4) & 7.45 & 0.29~(3) & 7.33 & 0.05~(2) & 7.27 & 0.13~(3) & 7.33 & 0.17 & \multicolumn{2}{c}{\ldots} & {\em 7.42} & {\em 0.23} & {\em 7.19} & {\em 0.21} & \multicolumn{2}{c}{\ldots}\\
\ion{Si}{iii} & 7.56 & 0.01 & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & 8.13 & 0.23~(3) & \multicolumn{2}{c}{\ldots} & {\em 7.26} & {\em 0.19} & {\em 7.42} & {\em 0.23} & {\em 7.19} & {\em 0.21} & 7.50 & 0.15\\
\ion{P}{ii} & 5.56 & 0.06 & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & 5.53 & 0.06~(4) & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots}\\
\ion{S}{ii} & 7.20 & 0.06 & {\em 7.15} & {\em 0.07}~(14) & {\em 7.07}& {\em 0.08}~(10) & {\em 7.12} & {\em 0.08}~(16) & {\em 7.05} & {\em 0.09}~(26) & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots}\\
\ion{S}{iii} & 7.20 & 0.06 & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & {\em 7.08} & {\em 0.04}~(2) & \multicolumn{2}{c}{\ldots} & {\em 7.24} & {\em 0.14} & \multicolumn{2}{c}{\ldots} & 6.87 & 0.26 & \multicolumn{2}{c}{\ldots}\\
\ion{Ca}{ii} & 6.35 & 0.01 & \multicolumn{2}{c}{6.31~(1)} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots}\\
\ion{Sc}{ii} & 3.10 & 0.01 & 2.57 & 0.14~(3) & 2.64 & 0.27~(2) & \multicolumn{2}{c}{2.42~(1)} & \multicolumn{2}{c}{\ldots} & 3.13 & 0.20 & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots}\\
\ion{Ti}{ii} & 4.94 & 0.02 & {\em 4.89} & {\em 0.13}~(29) & {\em 4.86} & {\em 0.12}~(25) & {\em 4.87} & {\em 0.09}~(15) & {\em 5.01} & {\em 0.07}~(4) & 4.86 & 0.25 & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots}\\
\ion{V}{ii} & 4.02 & 0.02 & 3.57 & 0.06~(6) & 3.63 & 0.08~(6) & 3.55 & 0.04~(3) & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots}\\
\ion{Cr}{ii} & 5.69 & 0.01 & 5.62 & 0.08~(29) & 5.55 & 0.08~(24) & 5.28 & 0.08~(21) & 5.42 & 0.11~(8) & 5.61 & 0.23 & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots}\\
\ion{Mn}{ii} & 5.53 & 0.01 & 5.38 & 0.02~(7) & 5.36 & 0.04~(6) & \multicolumn{2}{c}{5.24~(1)} & \multicolumn{2}{c}{5.33~(1)} & 5.81 & 0.20 & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots}\\
\ion{Fe}{i} & 7.50 & 0.01 & 7.34 & 0.12~(21) & 7.37 & 0.10~(13) & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & 7.56 & 0.24 & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots}\\
\ion{Fe}{ii} & 7.50 & 0.01 & {\em 7.52} & {\em 0.09}~(35) & {\em 7.42} & {\em 0.05}~(35) & {\em 7.34} & {\em 0.07}~(24) & {\em 7.47} & {\em 0.08}~(20) & 7.40 & 0.11 & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots}\\
\ion{Fe}{iii} & 7.50 & 0.01 & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & 7.45 & 0.07~(10) & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} &  7.36 & 0.20 & \multicolumn{2}{c}{\ldots}\\
\ion{Ni}{ii} & 6.25 & 0.01 & 6.30 & 0.06~(7) & 6.17 & 0.04~(7) & 6.00 & 0.01~(2) & 5.91 & 0.08~(4) & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots}\\
\ion{Sr}{ii} & 2.92 & 0.02 & 2.37 & 0.04~(2) & 2.41 & 0.03~(2) & 2.49 & 0.04~(2) & \multicolumn{2}{c}{\ldots} & 2.41 & 0.21 & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots}\\
\ion{Ba}{ii} & 2.22 & 0.02 & \multicolumn{2}{c}{2.00~(1)} & \multicolumn{2}{c}{2.13~(1)} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots} & \multicolumn{2}{c}{\ldots}\\
\hline
\end{tabular}}
$$
$^{a}$ Grevesse \& Sauval (\cite{GrSa98}): meteoritic abundances for
non-volatile elements; \hspace{1mm}
$^{b}$ Venn (\cite{Venn95a}, \cite{Venn95b}); \hspace{1mm}
$^{c}$ Daflon \& Cunha (\cite{DaCu04}): mean values of stars from 17
clusters/associations at $R_0$~$\pm$2~kpc; \hspace{1mm}
$^{d}$ Gummersbach et~al. (\cite{Gummersbachetal98}): mean values of
10 stars at $R_0$~$\pm$2~kpc; \hspace{1mm}
$^{e}$ mean from Kilian (\cite{Kilian92}, \cite{Kilian94}); \hspace{1mm}
$^{f}$ Rolleston et~al. (\cite{Rollestonetal00}): mean of
stars from 11 clusters/associations at $R_0$~$\pm$2~kpc; 
$^{*}$~reanalysis by Venn \& Przybilla~(\cite{VePr03}) -- result from
Venn~(\cite{Venn95b}): ${\it 8.05}\pm{\it 0.19}$; \hspace{1mm}
$^{**}$ \ion{Ne}{ii}.
\end{table}